

Earth's Future

RESEARCH ARTICLE

10.1029/2022EF003407

Key Points:

- Integrating maps of migrations of indigenous-managed reindeer herds with remote sensing enabled analysis of large-scale herbivory impacts
- Areas on the Yamal Peninsula with higher reindeer herbivory pressure consistently had lower coverage of tall woody vegetation
- The productivity of tall woody vegetation was less sensitive to climate variability where reindeer density was higher

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

M. P. Spiegel, marcus.spiegel@ouce.ox.ac.uk

Citation:

Spiegel, M. P., Volkovitskiy, A., Terekhina, A., Forbes, B. C., Park, T., & Macias-Fauria, M. (2023). Top-down regulation by a reindeer herding system limits climate-driven Arctic vegetation change at a regional scale. *Earth's Future*, 11, e2022EF003407. https://doi. org/10.1029/2022EF003407

Received 2 DEC 2022 Accepted 25 MAY 2023

Author Contributions:

Conceptualization: Marcus P. Spiegel, Marc Macias-Fauria Data curation: Alexander Volkovitskiy, Alexandra Terekhina, Taejin Park Formal analysis: Marcus P. Spiegel, Bruce C. Forbes Investigation: Alexander Volkovitskiy, Alexandra Terekhina Methodology: Marcus P. Spiegel, Alexander Volkovitskiy, Alexandra Terekhina, Marc Macias-Fauria Supervision: Marc Macias-Fauria

© 2023 The Authors. Earth's Future published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Top-Down Regulation by a Reindeer Herding System Limits Climate-Driven Arctic Vegetation Change at a Regional Scale

Marcus P. Spiegel¹, Alexander Volkovitskiy², Alexandra Terekhina², Bruce C. Forbes³, Taejin Park^{4,5}, and Marc Macias-Fauria¹

¹Biogeosciences Lab, School of Geography and the Environment, University of Oxford, Oxford, UK, ²Arctic Research Station of Institute of Plant and Animal Ecology, Ural Branch, Russian Academy of Sciences, Labytnangi, Russia, ³Arctic Centre, University of Lapland, Rovaniemi, Finland, ⁴NASA Ames Research Center, Moffett Field, CA, USA, ⁵Bay Area Environmental Research Institute, Moffett Field, CA, USA

Abstract Warming-driven growth of tall woody vegetation in the Arctic has the potential to accelerate climate change through multiple positive feedbacks. Local-scale evidence suggests that large herbivores limit this vegetation shift, but there is uncertainty at larger, regional scales whether current herbivory pressure is a major top-down control on ecosystem structure and functioning. Across a 67,000 km² region of the Yamal Peninsula in West Siberia, we integrated satellite remote sensing with a novel data set mapping the migrations of herds comprising 151,000 domesticated reindeer. Where reindeer numbers varied over space, higher reindeer herbivory pressure was consistently linked with lower coverage of tall woody vegetation. Within areas dominated by this vegetation type, productivity and climate were increasingly decoupled where reindeer density was higher. Our spaceborne fingerprint detection suggests that large herbivores, at current population densities, counteract Arctic vegetation responses to climate change over large spatial scales.

Plain Language Summary Large herbivores influence tundra vegetation via their grazing,

trampling, and nutrient inputs from feces and urine. However, the extent of these impacts remains unclear as we seek to understand and predict the responses of Arctic ecosystems to climate change. Using maps that we developed that record the annual summertime movements of herds comprising 151,000 domesticated reindeer managed by indigenous Nenets herders, we assessed whether reindeer had an effect on vegetation productivity and composition across a 67,000 km² region of the Yamal Peninsula in West Siberia, an area about the size of the Republic of Ireland. Analyses of satellite data comparing areas with varying reindeer density consistently showed that higher reindeer herbivory pressure was limiting the coverage of tall woody vegetation and its expansion in response to warming. This effect had not been observed previously at such a large spatial scale and was significant enough to be detected from space. Our findings demonstrate that reindeer have a substantial impact on ecosystem structure and functioning, which likely has implications for the carbon cycle and energy balance within the Arctic and beyond.

1. Introduction

Large herbivores have significant impacts on ecosystem structure and functioning throughout Earth's biomes (Enquist et al., 2020; Kristensen et al., 2022; Lundgren et al., 2020). During the Pleistocene, land in the non-glaciated high latitudes supported a diverse assemblage of large herbivores, including mammoth, woolly rhino, and wild horse, that grazed on a productive graminoid- and forb-rich heterogeneous vegetation matrix and likely maintained it through selective foraging, increased nutrient cycling, bark-stripping, and trampling (Macias-Fauria et al., 2020; Olofsson & Post, 2018; Zimov et al., 2012). In the modern moss- and shrub-dominated Arctic tundra, reindeer (*Rangifer tarandus* L.) are the only remaining large herbivores still widespread (Olofsson & Post, 2018). Local-scale studies, utilizing variations in grazing intensity due to the introduction of reindeer on islands (Klein, 1968; Ricca et al., 2016), different grazing regimes separated by management fences (Egelkraut et al., 2018; Olofsson et al., 2001, 2004; Ylänne et al., 2018), and exclosure experiments (Cahoon et al., 2012; Kolari et al., 2019; Metcalfe & Olofsson, 2015; Ylänne et al., 2020), consistently have found that reindeer cause declines in deciduous shrubs. These woody plants, especially willow and birch species, are highly palatable to reindeer, and are not as tolerant to repeated defoliation and trampling as graminoids (van der Wal, 2006). Thus, reindeer herbivory often drives transitions from shrub- to graminoid-dominated vegetation (Olofsson & Post, 2018).



Spiegel

Writing - original draft: Marcus P.

Writing - review & editing: Marcus

P. Spiegel, Alexander Volkovitskiy, Alexandra Terekhina, Bruce C. Forbes,

Taejin Park, Marc Macias-Fauria

The expansion of woody vegetation, especially tall shrubs, with climate change in the Arctic is well-documented (Myers-Smith et al., 2011). Studies have shown that, under observed and experimental warming, reindeer herbivory counteracts this vegetation shift, weakening or even negating the positive relationship between annual mean temperature and shrub growth and preventing the shrubline from advancing (Bråthen et al., 2017; Olofsson & Post, 2018; Olofsson et al., 2009; Post & Pedersen, 2008; Väisänen et al., 2014). Since the transition to more shrub-dominated tundra leads to lower albedo and increased permafrost thaw (Parker et al., 2021; Sturm et al., 2005), reindeer herbivory in the rapidly warming Arctic may have important Earth system implications by mitigating these positive climate change feedbacks.

Despite the strong evidence that reindeer herbivory acts as a significant top-down control when reindeer are concentrated over small areas, there is uncertainty over whether reindeer, at current population densities, have a substantial impact at larger spatial scales across the Arctic. Remote sensing is emerging as a tool for analyzing Arctic herbivory impacts, and has been used, for example, to detect a reduction in the normalized difference vegetation index (NDVI), a spectral reflectance metric for vegetation "greenness" (Raynolds et al., 2012), over a 770 km² area in the year after vole and lemming peaks (Olofsson et al., 2012). Several studies have found that reindeer reduce NDVI on summer pastures, but these effects were typically weak or inconsistent when comparing across differences in herd sizes or animal densities over space (Campeau et al., 2019; Newton et al., 2014; Rickbeil et al., 2015). Conversely, a study of migratory tundra caribou (same species as reindeer) herds in North America found that an increase in summer NDVI led to declines in caribou populations, likely related to the expansion of shrub species with strong antibrowsing defenses (Fauchald et al., 2017). Their results showed no evidence of a top-down effect of caribou on the vegetation, but instead implied a bottom-up effect initiated by a warming-driven increase in plant biomass. Although testing the relationship between reindeer abundances and NDVI provides an indication of how reindeer grazing is linked with vegetation productivity, analyses relying on NDVI alone cannot detect differences in vegetation cover, the main impact of herbivory found in many field studies. A lack of spatially explicit reindeer data has thus far hindered large-scale comparisons of vegetation cover across variations in grazing pressure over space or time (e.g., Rees et al., 2003; Verma et al., 2020; Walker et al., 2009).

Here, we analyze the influence of reindeer on vegetation on the Yamal Peninsula in West Siberia, where indigenous Nenets nomads manage the world's largest domesticated reindeer herds (~225,000 animals in total) and migrate up to 1,200 km annually (Forbes et al., 2009; Stammler, 2005). Local observations on Yamal have found that reindeer promote a transformation from shrub- to graminoid-dominated tundra and linked higher grazing intensity to decreased height and growth of willow shrubs (Forbes et al., 2009; Pajunen, 2009). The interactions between reindeer and vegetation in this system are dictated by the style of reindeer herding practised by the tundra Nenets, which affects how the reindeer move across and use the landscape. As described in the Russian literature, this herding style belongs to the so-called Samoyedic type—a tradition common in the tundra zone of Europe and West Siberia, where it was also adopted by several other ethnic groups, including the Khanty and Komi-Izhma (Habeck, 2007; Istomin & Dwyer, 2021; Jernsletten & Klokov, 2002). The defining features of Samoyedic reindeer herding, common to all the local variants, are the large herds of up to several thousand reindeer, extended migrations that enable the rotation of seasonal pastures, and herders' constant control over their herds (i.e., 24 hr a day, 7 days a week). The greatest attention is required during the summer, especially during periods of intense bloodsucking insect harassment in July. The emphasis on controlling animals dictates the nomadic lifestyle of the reindeer pastoralists of the Yamal Peninsula, who live year-round in the tundra. Thus, analysis of the influence of reindeer on the ecosystem of Yamal must incorporate an understanding of the role of humans as actors directing and modifying reindeer movements in response to multiple factors (ecological, weather, social, etc.) (Dwyer & Istomin, 2008). These features of herding in Yamal distinguish the reindeer herbivory patterns from those of both wild reindeer populations and free grazing semi-domestic reindeer in Fennoscandia.

Within the Yamal Peninsula, reindeer herding collectives are either private households (most local reindeer herders) or brigades of reindeer herding enterprises—the "heirs" of Soviet-era state-owned reindeer enterprises (*sovkhozy*). Both types of collectives have followed relatively stable patterns of seasonal movement, inherited largely from the routes of past generations of Nenets nomads, described since the 19th century (Evladov, 1992). Dozens of households migrate along the same "corridors" with consistent movement timings from year-to-year. Although the winter pasture locations of a certain Nenets household may vary within their corridor, the nomadic route and pasture areas visited in the summer months are more stable. In fact, the local identity of most Nenets is defined mainly by their summer pastures (Volkovitskiy & Terekhina, 2020).



Figure 1. Reindeer herds and their summer migrations on the Yar-Sale and Panayevsk tundra. The majority of Yamal reindeer have summer pastures on the southern and central parts of the Peninsula, on the Yar-Sale and Panayevsk tundra, whereas the Taz Peninsula to the east has many fewer reindeer. The locations of towns and villages serving as reindeer herding centers, Panayevsk, Yar-Sale, Seyakha, and Nyda, are also shown. We present a new data set of polygons depicting the herd locations in June, July, and August, colored to represent the number of reindeer in each herd in 2020, for both the private and enterprise herds with summer pastures on the Yar-Sale and Panayevsk tundra. The imagery base map is from Esri and Earthstar Geographics, and the water bodies base map is derived from the MOD44W Land Water Mask product (Carroll et al., 2017).

Given the intensity of reindeer herbivory on the Yamal Peninsula and the long-term stability of the seasonal migration patterns of the herds, we investigated whether variations in vegetation productivity and composition over space were linked to differences in the timing and density of reindeer pressure. Central to our analysis is a new data set we present that maps the annual June-August migrations of all the herds on the southern and central parts of the peninsula encompassing the Yar-Sale and Panayevsk tundra (the territory of the Yarsalinskii and Panayevskii *sovkhozy*, Figure 1), which has the highest summer density of reindeer on Yamal. Prior depictions of herd migrations on the Yamal Peninsula have either focused on the specific routes of a small subset of herds or the general movement patterns within the different *sovkhozy* (Degteva & Nellemann, 2013; Golovnev et al., 2014; Stammler, 2005), making our data set the first comprehensive map of reindeer numbers and summer ranges on an extensive region of 67,000 km² covering more than half of the Yamal Peninsula.

Our detailed, spatially explicit data set provides the opportunity to test hypotheses linking reindeer herbivory to vegetation responses at the regional scale. Alternative ecological approaches involving experimental manipulations are not feasible at such broad scales, and process-based models similar to those used in climate change attribution studies are rarely appropriate for biotic systems due to the need to incorporate myriad extrinsic factors with complex interactions (Parmesan et al., 2013). Thus, our approach adopts a framework along the lines of "associative pattern attribution," a method that evaluates whether multiple lines of evidence point to a coherent and parsimonious conclusion in accordance with a predicted biotic response (Hegerl et al., 2009). Our lines of evidence are derived from comparisons over space between areas with varying reindeer pressure. The analysis consists of three separate spatial comparisons: (a) between the Yar-Sale/Panayevsk tundra and the neighboring Taz Peninsula, which has considerably fewer reindeer but otherwise resembles Yar-Sale/Panayevsk geographically, geologically, and climatically (see Text S1 in Supporting Information S1 for a detailed comparison); (b) between areas in Yar-Sale/Panayevsk that are within summer herd ranges and areas outside of the migrations; and (c) within the summer ranges, considering differences in the density of animals on different pasture land. In all comparisons, we controlled for landscape position by dividing the areas into upland, slope, and lowland landform units (see Section 2, Figure S2 in Supporting Information S1) and separated the grazing areas within Yar-Sale/Panayevsk depending on the month in which they were used (Figure 1).

We formulated two main hypotheses informed by observations of reindeer impacts at the local scale that we tested at the regional scale using each of the three comparisons. First, we hypothesized that reindeer herbivory would alter the distribution of landcover types, specifically linking lower coverage of tall woody vegetation with higher reindeer numbers. Our second hypothesis was that vegetation productivity in areas with higher reindeer numbers would be less responsive to climate variability, measured by the vegetation sensitivity index (VSI, after Seddon et al. (2016)). Herbivory in the Arctic tends to have the strongest effect on the species expected to be favored by warming (Olofsson & Post, 2018), so we anticipated that areas with higher herbivory pressure would have lower VSI, indicating that reindeer were constraining vegetation responses to climate change. Whereas differences in the proportions of landcover types present can be interpreted as the cumulative impacts of reindeer herbivory over time, differences in the VSI represent herbivory's effects on the responsiveness of vegetation to recent climate variability. Within each of the comparisons testing our main hypotheses, we expected landscape position and seasonal timing to mediate the magnitude and the significance of the vegetation responses to variation in reindeer pressure. Despite the notably flat terrain of Yamal and Taz, which is mostly less than 70 m above sea level, landform is an important determinant of both vegetation cover in the region (Walker et al., 2009) and the amount of time reindeer spend on a given area. We hypothesized that differences due to herbivory would be greatest in the uplands, which are preferred by herders for migrating and campsites because they provide greater insect relief for the reindeer, dry ground for the campsite, and an elevated view of the landscape (Degteva & Nellemann, 2013; Skarin et al., 2020). We also hypothesized that the pastures used earlier in the growing season would exhibit stronger responses to herbivory. Reindeer tend to prefer deciduous shrubs for their diet in the early growing season, at which point they are most vulnerable to herbivory, before the amount of nitrogen and phosphorus in their leaves declines rapidly at the end of June (Chapin III, 1980; Chapin III et al., 1980). By comparing these hypothesized patterns to remote sensing observations, we can address whether the spatial variation in vegetation composition and productivity over the vast 67,000 km² Yar-Sale/Panayevsk tundra in Yamal has the fingerprints of reindeer herbivory.

2. Materials and Methods

2.1. Study Area

The study area is located within the Yamal Peninsula and neighboring Taz Peninsula in the Yamal-Nenets Autonomous Okrug, West Siberia, Russia (Figure 1). The area within the Yamal Peninsula was confined to the territory of the Yar-Sale/Panayevsk tundra, where herd migrations are typically long and linear compared with the more circular migrations in the northern Seyakha tundra (Degteva & Nellemann, 2013). The area on the Taz Peninsula was delineated based on the extent included in the Circumpolar Arctic Vegetation Map (Walker et al., 2005).

A detailed comparison of the Yar-Sale/Panayevsk tundra and the Taz Peninsula based on topography (Porter et al., 2018), geology (Drozdov et al., 2005), and temperature (Karger et al., 2017, 2018) is presented in Text S1 in Supporting Information S1. Taz was chosen as a region for comparison with Yar-Sale/Panayevsk that differs mainly in the abundance of reindeer while having a similar physical environment. Overall, the whole study area is low and flat and traverses primarily bioclimate subzones D and E, although the northwest extent of Yar-Sale/Panayevsk is in subzone C (Walker et al., 2005). The land was unglaciated during the last glaciation and consists of sandy and clayey marine, alluvial, and lacustrine sediments (Walker et al., 2009). The region is underlain by continuous, ice-rich permafrost, and the vegetation consists of a mosaic of tall woody shrub tundra (dominated by *Salix* spp., *Betula*), low stature shrub tundra (*Betula*, ericoids), and mires dominated by sedges (*Carex* spp., *Eriophorum* spp.) (Botch et al., 1971; Forbes et al., 2009). The geomorphology is shaped by big meandering rivers, thermokarst lakes, and frequent cryogenic landslides (Walker et al., 2009).

2.2. Mapping of Reindeer Herds and Migrations

We recorded the number of reindeer in each herd on Yar-Sale/Panayevsk and mapped the extent of the area on which they graze annually in June, July, and August using official statistics, semi-structured interviews, and participatory mapping with indigenous experts. The official statistics for the Yar-Sale herding enterprise were based on corral (round-up) calculations of animals in the enterprise herds in 2019–2020. Interview participants included "zootechnicians," experts who are consulted on reindeer herding matters such as breeding and seasonal movements, from the Yar-Sale and Panayevsk herding enterprises as well as employees of the Kharp *obshchina*, a community structure facilitating the cooperation of private reindeer herders. The Kharp *obshchina* is the largest in Yamal, accounting for the ownership of more than 60,000 reindeer. All applicable international, national and/ or institutional guidelines for the social research were followed. Interviewees consented to be interviewed and have approved the information and the sharing of it publicly.

We used printed GGTS (State GIS Center) maps (1:200,000) to mark the herd locations during the interviews. After separately recording the reindeer numbers for each herd, we checked their totals against the 2019–2020 reconstructed official estimate of all reindeer on Yar-Sale/Panayevsk (155,000–160,000 animals). We counted 37,747 reindeer in Yar-Sale brigades, 19,700 in Panayevsk brigades, and 94,000 in private herds, giving 151,447 total reindeer. Since we mapped all major herds, we presume that the reindeer constituting the small discrepancy between our count and the reconstructed estimate are spread over the whole region.

After digitizing the mapped routes into polygons for each herd's location in each month, we created monthly maps giving the number of reindeer across the Yar-Sale/Panayevsk tundra. Animal densities were calculated by dividing the number of reindeer by the amount of land area (excluding water bodies) within their respective range for each month based on the 20 m landcover classification of Bartsch et al. (2019). We dealt with overlapping polygons by adding the reindeer numbers and densities, but since this inflated the amount of reindeer pressure occurring within small intersecting areas at the edge of the ranges of multiple herds, we discarded any new polygons smaller than 10 km^2 (<2% of the total pasture area in total) in subsequent analyses in which reindeer density was considered.

Many fewer reindeer graze on the Taz Peninsula compared with Yar-Sale/Panayevsk. The most recent published information on reindeer herds grazing in the summer on Taz within our study area estimated their numbers at \sim 10,000, although this is reflective of the 1980–1990's (Kvashnin, 2009). Our conversations with local people in Nyda, the reindeer herding center in the region, confirmed that the number of reindeer in Taz is significantly lower than in any part of the Yamal Peninsula.

2.3. Landcover Classification

The mapping of landcover types used the landcover classification of Bartsch et al. (2019) (Figure S1c in Supporting Information S1). This landcover product was developed using a supervised classification of Sentinel-1 and Sentinel-2 imagery from 2016 to 2018 and groundtruthed by local experts. It contains 18 terrestrial classes and three aquatic classes. We merged the terrestrial classes into six broad classes, including "sparse vegetation," "low stature shrub tundra," "tall woody vegetation," "meadow," "inundated/landslides," and "barren/rare vegetation." The associations between the original class definitions and the landcover classes we used in our analysis are presented in Table S1 in Supporting Information S1. To facilitate our adaptation of the landcover classes, we compared the Bartsch et al. (2019) landcover map with the 1 km resolution Circumpolar Arctic Vegetation Map (Raynolds et al., 2019). This comparison helped us determine, for example, which classes aligned with our definition of tall shrubs as exceeding 0.4 m height.

"Low stature shrub tundra" includes areas with a mixture of graminoids, forbs, erect dwarf shrubs (0.1–0.4 m), and prostrate dwarf shrubs (<0.1 m) (Myers-Smith et al., 2011). This class is distinguished from "tall woody vegetation," which consists of tall shrub tundra as well as deciduous, coniferous, and mixed forest, although tree-dominated pixels cover less than 4% of the study area and are exclusively concentrated in the far south of the area. The "sparse vegetation" class describes land on mostly sandy soil with few shrubs and potentially dry biological crusts. The "inundated/landslide" class consists of areas within disturbed habitats that are seasonally inundated or within landslide scars. Landslide scars are commonly found on Yamal, and landslides are increasing in frequency with ongoing rapid permafrost thaw (Leibman, 1995; Walker et al., 2009). In our analyses, we discarded pixels in the "meadow" and "barren/rare vegetation" classes because they represented no more than 8% of the total pixels within any comparison. The meadow pixels were primarily confined to riparian areas rather than occurring throughout the landscape where they might be indicative of herbivory-driven transitions from moss- and shrub-dominated tundra to grasslands (Forbes et al., 2009).

2.4. Landform Delineation

Our landform classification was performed by applying a modified version of the Geomorphons algorithm (Jasiewicz & Stepinski, 2013) to the ArcticDEM v3.0 release (Porter et al., 2018) at 32 m resolution (Figure S2 in Supporting Information S1). The algorithm assigns each pixel to one of 10 common landform elements (e.g., ridge, pit) in a procedure that automatically chooses the most appropriate spatial scale for landform delineation at each location. We further divided the pixels assigned to the flat landform class into flat uplands and flat lowlands based on the gradient in their neighboring pixels, and then reclassified the remaining landform pixels into "uplands," "slopes," and "lowlands."

2.5. Vegetation Sensitivity Index (VSI)

We adapted the VSI from Seddon et al. (2016) to quantify the sensitivity of vegetation productivity to variability in temperature, cloudiness, and water availability. We computed the VSI over the study area at 1 km resolution using monthly time series of remote sensing data from 2000 to 2019 (Figure S1a in Supporting Information S1). The model used the moderate-resolution imaging spectroradiometer (MODIS) enhanced vegetation index (EVI) MOD13A2 version 6 product (Didan, 2015) as the response variable representing vegetation productivity. The three climate predictor variables were monthly maximum temperature (obtained from CHELSA version 2.1 (Karger et al., 2017, 2018)), ratio of actual to potential evapotranspiration (AET/PET) for water availability (obtained from MODIS MOD16A2GF version 6 (Running et al., 2019)), and cloudiness fraction (obtained from the MODIS MOD35L2 version 6.1 cloud mask product (Ackerman, 2015)). All time series were transformed to z-score anomalies using monthly means and standard deviations, and the months were filtered based on a minimum EVI (0.1) and temperature threshold (0°C) to identify the growing season for each pixel. Principal component regressions were used to identify the relative importance of the predictor variables in driving variations in EVI for each pixel, and these coefficients were multiplied by the log₁₀-transformed ratios of EVI variability and variability in each climate time series to arrive at sensitivity. The sensitivity (VSI) is a relative metric, scaled from 0 to 100 over the study area, and was mapped in the MODIS sinusoidal projection.

2.6. Statistical Analyses

To address our first hypothesis, that reindeer had influenced landcover, we analyzed the variation in landcover distribution with herbivory pressure. We compared the proportions of pixels in the four selected landcover classes between regions (Yar-Sale/Panayevsk and Taz) and between different pasture areas within Yar-Sale/Panayevsk by calculating odds ratios. These comparisons were performed separately for pixels within "upland," "slope," or "lowland" landforms. We used log-linear modeling to test whether reindeer pressure (categorically differentiated by region or by pasture area), landform, and the interactions between them significantly affected the proportions of sparse vegetation and tall woody vegetation pixels. We tested a suite of possible models each representing different hypotheses regarding which factors were significant predictors using the glm function from the stats package (R Core Team, 2020) with a Poisson variance and log link function. We evaluated the fit of each model using the residual deviance statistic G^2 (von Eye et al., 2012).

For our analysis of landcover across variations in reindeer density within the area of Yar-Sale/Panayevsk used as summer pastures, we quantified landcover anomalies as the difference from the expected ratio of sparse vegetation to tall woody vegetation pixels. The ratio was calculated as the fraction of sparse vegetation pixels, (sparse/(sparse + tall woody)), and the expected ratio was based on the landforms present, calculated as a weighted average combining the number of pixels and the mean sparse fraction for each landform over the whole pertinent area.

To test whether reindeer herbivory had affected the sensitivity of vegetation productivity to climate as suggested by our second hypothesis, we analyzed variations in VSI with reindeer pressure using our three comparisons. The VSI map, at 1 km resolution, is much coarser than the landcover (20 m) and landform classifications (32 m). To enable analysis at a common spatial scale, each VSI pixel was assigned to a landcover class if that class covered at least half of the 1 km pixel and to a landform class if it covered at least 2/3 of the pixel. The higher threshold for landforms was chosen because of the fewer landform classes and to avoid potentially complex dynamics in more rugged terrain. Pixels that did not meet these thresholds were not used in the subsequent comparisons, but the classified pixels were well-distributed over the area with ample coverage (Figure S3 in Supporting Information S1).

Comparisons of VSI in which only two samples were compared (i.e., between Yar-Sale/Panayevsk and Taz within a common landform or landcover) used the Wilcoxon rank sum test (wilcox.test from stats (R Core Team, 2020)) to determine significance. Comparisons involving more than two samples or multiple categorical variables used one-way ANOVAs or multi-way ANOVAs with type II sum of squares (anova test from rstatix (Kassambara, 2020)). When the ANOVAs found significant differences, we evaluated the differences between pairs of samples using Tukey's test (TukeyHSD from stats (R Core Team, 2020)). To quantify the relationship between VSI and reindeer density within pasture areas, we performed linear regressions between these two variables (Im from stats (R Core Team, 2020)).



Earth's Future



Figure 2. Distribution of 20 m resolution landcover pixels based on the classification of Bartsch et al. (2019). The distributions are compared by region (a, c, and e) and by the different pasture areas used in the summer months by herds in the Yar-Sale/Panayevsk tundra (b, d, and f). "No pastures" represents areas without any known reindeer herds. The pixels are also divided topographically into landforms: (a, b) uplands, (c, d) slopes, and (e, f) lowlands. The percentage of pixels within each landcover type is written inside the bar segments, and the total number of pixels in the specified area is given above the bar. Note that for each landform class, the sum of the number of pixels in the different pasture areas (b, d, and f) exceeds the number of pixels in the Yar-Sale/Panayevsk region (a, c, and e) because pixels may belong to multiple pasture areas (i.e., grazed in multiple months).

3. Results

3.1. Variation in Landcover With Reindeer Herbivory Pressure

In comparisons of the relative proportions of landcover types between the Yar-Sale/Panayevsk tundra, where reindeer numbers are highest on Yamal, and the Taz Peninsula, where there are considerably fewer reindeer (Figures 2a, 2c, and 2e), and between areas with and without summer pastures within Yar-Sale/Panayevsk (Figures 2b, 2d, and 2f), the areas with higher reindeer pressure consistently had higher coverage of sparse vegetation and lower coverage of tall woody vegetation than areas with fewer reindeer. This pattern occurred within all landform units, but in accordance with our hypothesis, the effect was greatest in the uplands, where reindeer spend the most time. Based on odds ratios, the ratio of tall woody vegetation area to sparse vegetation area was 4.0 times lower on the uplands, 3.6 times lower on the slopes, and 3.2 times lower on the lowlands of Yar-Sale/Panayevsk compared to Taz.

Also in accordance with our expectations, the pastures grazed in June, when deciduous shrubs are most vulnerable to herbivory, generally had the most sparse vegetation area and least tall woody vegetation within Yar-Sale/Panayevsk. This result would be surprising in the absence of herbivory, as especially for the enterprise herds, these pastures tend to be located farther south in the low Arctic (Figure 1), where greater coverage of tall shrubs is expected (Myers-Smith et al., 2015). When comparing the June pastures to the no pastures area, the ratio

Deviance Statistics for Log-Linear Models				
Comparison	Assumption	Predictors	df	G^2
Regions	Joint independence	None	5	3,798,785
	Conditional independence	Only region	4	1,334,717
	Conditional independence	Only landform	3	2,533,505
	Homogenous association	Region and landform	2	9,034.8
Pasture areas	Joint independence	None	11	2,062,131
	Conditional independence	Only pasture area	8	1,782,651
	Conditional independence	Only landform	9	383,104
	Homogeneous association	Pasture area and landform	6	95,499

Table 1

Deviance Statistics for Log-Linear Models

Note. The suite of models evaluates possible explanations for the differences in the proportions of tall woody vegetation and sparse vegetation landcover pixels (see Figure 2). One set of models evaluates differences in landcover between Yar-Sale/ Panayevsk and Taz ("Regions"), while the other set evaluates differences between June pastures, July pastures, August pastures, and land not used for summer pastures in Yar-Sale/Panayevsk ("Pasture Areas"). For each model, df represents the degrees of freedom and G^2 gives the residual deviance comparing the fitted model against the saturated model with all possible effects including interactions.

of tall woody vegetation area to sparse vegetation area was 2.3 times lower on the uplands, 1.5 times lower on the slopes, and 1.2 times lower on the lowlands. Using a suite of log-linear models formulated to test the dependence of the proportions of landcover classes on categorical predictors, we found that both landforms and comparison areas (region in the first case, Yar-Sale/Panayevsk pasture areas in the second) significantly improved the model fit when their main effects and the interaction effects between them were included (see Table 1 for deviance statistics).

While there were also differences in the relative proportions of low stature shrub tundra and inundated/landslide area, they showed mixed responses to reindeer presence. These differences likely were not caused by variation in herbivory pressure, but potentially due to a higher density of rivers and lakes in Yar-Sale/Panayevsk (11.7% area covered by water, compared to 4.8% in Taz, calculated from the landcover classification), leading to greater coverage of inundated/landslide area than in Taz, and a tendency for reindeer herds to avoid landslide and mire areas (Skarin et al., 2020), resulting in lower coverage of this landcover in pasture areas within Yar-Sale/Panayevsk.

We also found that higher reindeer pressure remained linked to less tall woody vegetation and more sparse vegetation when considering variations in animal density across the pasture areas. Within each herd area of a given reindeer density, we calculated how the observed proportions of sparse vegetation and tall woody vegetation differed from the expectation based on the average coverage proportions associated with the landforms present (Figure 3). A positive value for this difference implies that the area contains more sparse vegetation and less tall woody vegetation than expected. The relationship between reindeer density and this difference was highly nonlinear, as the most pronounced effect occurred once the reindeer density exceeded a high threshold: 21 reindeer/km² on June pastures and 47.5 reindeer-months/km² when integrating over June–August. In all comparisons, points were mostly absent from the lower right of the plot area, representing high animal density and more tall woody vegetation than expected based on the landforms present. This tendency is highlighted by the contours, which display the results of a 2D kernel density estimation. Especially when only considering the uplands (Figures 3b and 3d), the outermost contour lines slope upwards (from lower left to upper right in the plot), indicating a similar probability of pasture areas having low reindeer density with more tall woody vegetation than expected and high reindeer density with more sparse vegetation than expected.

3.2. Effect of Reindeer Herbivory on Vegetation Sensitivity to Climate

Contrary to our second hypothesis, VSI values were consistently greater in Yar-Sale/Panayevsk than in Taz, albeit by a small margin, despite the much greater herbivory pressure in Yar-Sale/Panayevsk (Figure 4). Paired comparisons of the VSI between regions using the Wilcoxon rank sum test indicated a significant difference for all comparisons within landforms and all comparisons within landcover, except within tall woody vegetation.





Figure 3. Link between reindeer density and the amount by which the ratio of sparse vegetation to tall woody vegetation differs from the ratio that would be expected based on the distribution of landforms in each herd area. Each point represents a continuous herd area with a given reindeer density in (a, b) June or density summed over (c, d) June, July, and August. The difference from the expected ratio was calculated as (sparse fraction – expected sparse fraction), where the sparse fraction is (sparse/(sparse + tall woody)) and expected sparse fraction is a weighted average combining the number of pixels and the mean sparse fraction in each relevant landform class over all June pasture areas (a, b) or all summer pasture areas (c, d) in Yar-Sale/Panayevsk. The number of sparse vegetation and tall woody vegetation pixels on all landforms is included in panels (a, c), while the sparse fractions are only calculated on upland pixels in panels (b, d). The contour lines and shading display the results of 2D kernel density estimation.

When dividing Yar-Sale/Panayevsk into pasture areas, the VSI in each area remained slightly higher than in Taz within every landcover type except, notably, within tall woody vegetation. Whereas the mean VSI in tall woody vegetation still was higher in the no pasture area of Yar-Sale/Panayevsk than in Taz (+2.59), all reindeer pasture areas had a lower mean VSI (June: -1.38, July: -1.64, and August: -0.533) (Figure 5a). A one-way ANOVA indicated significant differences in the mean VSI in the comparisons within tall woody vegetation (F = 4.728, p < 0.001), but Tukey's test showed a significant difference only in the comparison of no pasture area with July pastures (p = 0.002).

The impact of reindeer herbivory on the sensitivity of tall woody vegetation to climate is further supported by the negative correlation between VSI and reindeer density within this landcover (Figures 5b and 5c). The estimated slope of the linear regression was -0.708 VSI/(reindeer/km²) (p = 0.0415) for June pastures and -0.268 VSI/ (reindeer·months/km²) (p = 0.0155) for all pasture areas.

4. Discussion

Our results suggest that reindeer herbivory on Yamal constrains the response of tall woody vegetation to warming and limits shrub expansion, shaping the landscape to consist of less tall woody vegetation and more sparse vegetation than it would have without the influence of animals. Employing an associative pattern attribution approach,





🛑 Yar-Sale/Panayevsk 🖨 Taz Peninsula

Figure 4. Comparisons of vegetation sensitivity index values mediated by (a) landforms, (b) landcover, and (c–e) the interactions between them. Landcover classes are excluded in panel (d, e) where there were fewer than three representative pixels in both regions. The lower and upper box boundaries correspond to the 25th and 75th percentiles, respectively, and the line inside the box gives the median. The number of 1 km pixels in each comparison is given in the boxes, and the significance level of the differences based on the Wilcoxon rank sum test is shown above them. NS indicates not significant, p > 0.05; a single asterisk (*) indicates p < 0.05; double asterisks (**) indicate p < 0.01; and triple asterisks (***) indicate p < 0.001.

we evaluated six lines of evidence, which entailed using three spatial comparisons of herbivory pressure to test each of two hypotheses regarding the responses of landcover distributions and VSI to reindeer herbivory. The results of the landcover comparisons were consistent with our hypotheses-higher reindeer pressure was linked to lower coverage of tall woody vegetation, as well as more sparse vegetation, especially on uplands and early in the growing season. However, the higher vegetation sensitivity to climate observed in Yar-Sale/Panayevsk with respect to the Taz Peninsula was unexpected, was not explained by differences in landforms, general relief, landcover, or substrate, and occurred in both pastures and non-pastures (Figures 4 and 5, Figure S4 in Supporting Information S1). We speculate that this pattern might be driven by differences in growing season climate variability between both regions during the period 2000–2019 over which the VSI is computed. Yet, despite this general pattern in VSI, vegetation sensitivity within Yar-Sale/Panayevsk was consistently lower in the more heavily grazed areas exclusively in the tall woody vegetation. This finding suggests that reindeer herbivory does not have an extensive impact on vegetation sensitivity, but that it still has an observable effect in tall woody vegetation, where vegetation is otherwise most sensitive and where we would expect herbivory to counteract climate effects as drivers of vegetation change (Olofsson et al., 2009; Post & Pedersen, 2008). Therefore, after considering all the evidence simultaneously, top-down control from reindeer herbivory emerged as the most parsimonious, consistent explanation for the observed patterns. Our findings contrast with earlier regional-scale work concluding that bottom-up, climate-driven increases in unpalatable shrubs in North America contributed to declines in caribou abundance (Fauchald et al., 2017). However, our analysis benefitted from finer spatial grain-whereas Fauchald et al. (2017) used the GIMMS NDVI3g product at $1/12^{\circ}$ resolution (~8 km), our coarsest remote sensing product was at 1 km resolution, >60 times finer resolution—which enabled the pixel-based observations to better represent landscape-scale ecosystem structure and functioning. Additionally, the spatial comprehensiveness of the herd size and movement data provided a more accurate picture of the heterogeneity in herbivory pressure







Figure 5. Comparison of vegetation sensitivity index (VSI) values between different pasture areas in Yar-Sale/Panayevsk and Taz (a), and the relationship between VSI and reindeer density within tall woody vegetation in June pastures (b), and all pasture areas (c). See the caption of Figure 4 for the explanation of the boxes. Only significant differences based on Tukey's tests are annotated: a single asterisk (*) indicates p < 0.05; double asterisks (**) indicate p < 0.01; and triple asterisks (***) indicate p < 0.001. Each point in panels (b and c) represents a 1 km VSI pixel within tall woody vegetation with a given June reindeer density (b) or density summed over June, July, and August (c). The linear least squares fits have a significant negative slope ((b)–0.708 VSI/(reindeer/km²), (c) –0.268 VSI/(reindeer·months/km²), p < 0.05) and are displayed with shaded 95% confidence intervals.

across the region, avoiding the implicit assumption often found within prior studies, including those in Yamal (e.g., Verma et al., 2020; Veselkin et al., 2021), that reindeer and changes in their population sizes are spread evenly over a region. The integration of high-resolution remote sensing products with spatially explicit reindeer population data provided the opportunity for multiple tests of our hypotheses, allowing us to carefully examine the competing effects of climate and herbivory on the vegetation on Yamal at a regional scale. This pattern-based methodology could be adapted for future studies testing whether, or the degree to which, large-scale ecosystem responses can be attributed to the influence of animals.

The magnitude of the vegetation response to herbivory detected in this study was significant but relatively modest, observed as small percentage differences in VSI and landcover composition between the reindeer areas and their controls, and primarily in the areas with the strongest grazing pressure. This result evokes prior work which found that reindeer limited tall shrub expansion in Northern Norway but only when their densities exceeded approximately 5 animals/km² (Bråthen et al., 2017). The threshold we found (i.e., 21 reindeer/km² on June pastures, Figure 3) was higher than that observed in Northern Norway, but this disparity may be related to differences between the Nenets' reindeer herding in Yamal and the practices in Fennoscandia. Although Nenets herders are generally guided by the natural ecology of reindeer, at the regional scale, the seasonal choice of pastures and the timing of movements are determined entirely by herders. In the summer, Nenets on the Yar-Sale/Panayevsk tundra migrate every 3-5 days on average and even while a household remains in one place between movements, the herders try to direct reindeer to new areas as time passes. Thus, we hypothesize that a lower density of Fennoscandian free-grazing reindeer without constant human control could exert greater pressure on vegetation, especially the preferred shrub pastures in June, by remaining longer in certain areas. Within the Yamal Peninsula, the greater vegetation response we detected on June pastures may be not only due to the vulnerability of the vegetation to herbivory in the early growing season, but also because many households pass through the same areas as they migrate before dispersing to their July and August pastures. As a result, it would primarily be the June pastures that exceed the threshold of reindeer density needed to cause significant impacts on the vegetation, despite the characteristic high rates of movement.

Furthermore, detecting the impacts of herbivory in an uncontrolled setting is complicated by challenges such as rapid vegetation recovery blurring the signal of the herbivory impact (Ravolainen et al., 2011) and uncertainty in the precise positioning and timing of animal movements. Thus, the true ecological magnitude of the herbivory impacts on Yamal may exceed our findings, and given the 67,000 km² regional expanse over which we made our observations, there likely are implications for carbon storage and albedo relevant to Earth system functioning (Meredith et al., 2019).

Our observations represent a snapshot and do not inform whether the impact of reindeer on Yamal is changing over time nor the extent to which it may counteract climate-driven shrub growth in the future. Repeated monitoring of vegetation is needed to track and anticipate how the region will respond as the Arctic continues to warm. Our data on summer locations of reindeer herds and their migratory corridors reflect at least the last 30 years, but reindeer herding practices in the area are also changing, as herders adapt to infrastructure development for hydrocarbon extraction (Forbes et al., 2009) and climate change, which affects the timing of the freezing and thawing of rivers and has led to an increase in rain-on-snow events (Forbes et al., 2016, 2022). The latter cause the severe icing of pastures, leading to high reindeer mortality. There also have been calls from both biologists and administrators to reduce the number of reindeer on Yamal and restrict their migrations in response to a recurring argument that the land is overgrazed, based primarily on a decline in lichen availability (Golovnev, 2017; Volkovitskiy & Terekhina, 2020). While the definition and perception of overgrazing is a complex, interdisciplinary topic, our findings can inform these discussions by providing empirical insights into the impact reindeer are having on the vegetation over a regional scale.

The potential for rewilding the Arctic with large herbivores to reduce the abundance of shrubs and slow climate change feedbacks has already drawn considerable attention, but questions remain as to how effectively such projects will provide these ecosystem services (Macias-Fauria et al., 2020; Olofsson & Post, 2018). Some of this discussion arises from uncertainty over whether Pleistocene megafauna had a large top-down role in maintaining the graminoid- and forb-rich vegetation, or whether they were a more passive element within a trophic chain in which vegetation cover was determined by climate as a bottom-up driver (e.g., Monteath et al., 2021). Our results support that herbivores, at current population densities, affect vegetation cover in the modern Arctic at a regional and potentially biome-wide scale. The domesticated reindeer herds of the Nenets are the largest in the world today, whereas estimates of past megafauna population sizes are highly uncertain (Monteath et al., 2021). The density of large herbivores across the Arctic historically may have been similar or likely much higher than the current reindeer density on Yamal (Zimov et al., 2012), and the modern perception of this density as high may be a case of "shifting base-line syndrome" (Pauly, 1995) occurring over long periods of time. Thus, especially given the larger and more diverse Pleistocene herbivore assemblage (Olofsson & Post, 2018), it is likely that herbivory in the Arctic during the Pleistocene had even stronger impacts than what we have found here. Further work is necessary to translate the fingerprints of reindeer herbivory we detected into ecosystem variables and their quantitative estimates, but our findings demonstrate that top-down regulation by herbivory must be considered to understand how tundra ecosystems have responded, are responding, and will respond to climate change.

Data Availability Statement

The data used in this study are available in the Oxford University Research Archive via https://doi.org/10.5287/ bodleian:E2x6bnggk under a Creative Commons license (CC BY 4.0) (Spiegel et al., 2022b). V1.0.0 of the code (Spiegel et al., 2022a) used in the pre-processing of the data, analysis, and figure generation is preserved at https://doi.org/10.5281/zenodo.7387883, available under a GNU General Public License v3.0 and developed openly using R version 4.0.0 (R Core Team, 2020).

References

Ackerman, S. (2015). MODIS atmosphere L2 cloud mask product, NASA MODIS adaptive processing system. Goddard Space Flight Center. https://doi.org/10.5067/MODIS/MOD35_L2.061

Bartsch, A., Widhalm, B., Pointner, G., Ermokhina, K., Leibman, M., & Heim, B. (2019). Landcover derived from Sentinel-1 and Sentinel-2 satellite data (2015–2018) for subarctic and Arctic environments. *Pangaea*. https://doi.org/10.1594/PANGAEA.897916

Botch, M. C., Gerasimenko, T. V., & Tolchelnikov, Y. C. (1971). Mires of the Yamal Peninsula (in Russian with English summary). *Botanicheskii Zhurnal*, 56, 1421–1435.

Bråthen, K. A., Ravolainen, V. T., Stien, A., Tveraa, T., & Ims, R. A. (2017). Rangifer management controls a climate-sensitive tundra state transition. *Ecological Applications*, 27(8), 2416–2427. https://doi.org/10.1002/eap.1618

Acknowledgments

We thank the employees of the Yar-Sale herding enterprise and Kharp obshchina for their expert insights into modern Yamal reindeer herding. This work was supported by the Clarendon Fund (to M.P.S.): the Russian Ministry of Science and Higher Education program "Terrestrial ecosystem of Northwestern Siberia: assessment of the modern transformation of the communities," 122021000089-9 (to A.V. and A.T.): the Russian Science Foundation Project 22-24-00642 (to A.T.); the European Union's Horizon 2020 research and innovation programme under Grant 869471 (to M.P.S., B.C.F. and M.M.F.): Academy of Finland Decisions 256991 and JPI Climate 291581 (to B.C.F); and NASA Earth Science Directorate Grant NNX16AO34H (to T.P.).

- Cahoon, S. M. P., Sullivan, P. F., Post, E., & Welker, J. M. (2012). Large herbivores limit CO₂ uptake and suppress carbon cycle responses to warming in West Greenland. *Global Change Biology*, *18*(2), 469–479. https://doi.org/10.1111/j.1365-2486.2011.02528.x
- Campeau, A. B., Rickbeil, G. J. M., Coops, N. C., & Côté, S. D. (2019). Long-term changes in the primary productivity of migratory caribou (*Rangifer tarandus*) calving grounds and summer pasture on the Quebec-Labrador Peninsula (northeastern Canada): The mixed influences of climate change and caribou herbivory. *Polar Biology*, 42(5), 1005–1023. https://doi.org/10.1007/s00300-019-02492-6
- Carroll, M. L., DiMiceli, C. M., Wooten, M. R., Hubbard, A. B., Sohlberg, R. A., & Townshend, J. R. G. (2017). MOD44W MODIS/Terra land water mask derived from MODIS and SRTM L3 global 250m SIN grid V006 [Dataset]. NASA EOSDIS Land Processes DAAC. Retrieved from https://catalog.data.gov/dataset/modis-terra-land-water-mask-derived-from-modis-and-srtm-l3-global-250m-sin-grid-v006
- Chapin, F. S., III. (1980). Nutrient allocation and responses to defoliation in tundra plants. Arctic and Alpine Research, 12(4), 553. https://doi.org/10.2307/1550500
- Chapin, F. S., III, Johnson, D. A., & McKendrick, J. D. (1980). Seasonal movement of nutrients in plants of differing growth form in an Alaskan tundra ecosystem: Implications for herbivory. *Journal of Ecology*, 68(1), 189. https://doi.org/10.2307/2259251
- Degteva, A., & Nellemann, C. (2013). Nenets migration in the landscape: Impacts of industrial development in Yamal Peninsula, Russia. *Pastoralism*, 3(1), 1–21. https://doi.org/10.1186/2041-7136-3-15
- Didan, K. (2015). MOD13A3 MODIS/Terra vegetation indices monthly L3 global 1km SIN grid V006. [Dataset]. NASA EOSDIS Land Processes DAAC. https://doi.org/10.5067/MODIS/MOD13A3.006
- Drozdov, D. S., Rivkin, F. M., Rachold, V., Ananjeva-Malkova, G. V., Ivanova, N. V., Chehina, I. V., et al. (2005). Electronic atlas of the Russian Arctic coastal zone. *Geo-Marine Letters*, 25(2–3), 81–88. https://doi.org/10.1007/s00367-004-0189-7
- Dwyer, M. J., & Istomin, K. V. (2008). Theories of nomadic movement: A new theoretical approach for understanding the movement decisions of Nenets and Komi reindeer herders. *Human Ecology*, 36(4), 521–533. https://doi.org/10.1007/s10745-008-9169-2
- Egelkraut, D., Aronsson, K.-Å., Allard, A., Åkerholm, M., Stark, S., & Olofsson, J. (2018). Multiple feedbacks contribute to a centennial legacy of reindeer on tundra vegetation. *Ecosystems*, 21(8), 1545–1563. https://doi.org/10.1007/s10021-018-0239-z
- Enquist, B. J., Abraham, A. J., Harfoot, M. B. J., Malhi, Y., & Doughty, C. E. (2020). The megabiota are disproportionately important for biosphere functioning. *Nature Communications*, 11(1), 1–11. https://doi.org/10.1038/s41467-020-14369-y
- Evladov, V. P. (1992). Through the Yamal tundra to the Belyi Island. Expedition to the far North of the Yamal Peninsula in 1928–1929. Institute of the problems of Northern development, SB RAS. (in Russian).
- Fauchald, P., Park, T., Tømmervik, H., Myneni, R., & Hausner, V. H. (2017). Arctic greening from warming promotes declines in caribou populations. Science Advances, 3(4), e1601365. https://doi.org/10.1126/sciadv.1601365
- Forbes, B. C., Kumpula, T., Meschtyb, N., Laptander, R., Macias-Fauria, M., Zetterberg, P., et al. (2016). Sea ice, rain-on-snow and tundra reindeer nomadism in Arctic Russia. *Biology Letters*, 12(11), 20160466. https://doi.org/10.1098/rsbl.2016.0466
- Forbes, B. C., Kumpula, T., Meschtyb, N., Laptander, R., Macias-Fauria, M., Zetterberg, P., et al. (2022). Coping with a warming winter climate in Arctic Russia: Patterns of extreme weather affecting Nenets reindeer nomadism. In *Resilience through knowledge Co-production* (pp. 217–232). Cambridge University Press. https://doi.org/10.1017/9781108974349.017
- Forbes, B. C., Stammler, F., Kumpula, T., Meschtyb, N., Pajunen, A., & Kaarlejärvi, E. (2009). High resilience in the Yamal-Nenets social-ecological system, west Siberian Arctic, Russia. Proceedings of the National Academy of Sciences of the United States of America, 106(52), 22041–22048. https://doi.org/10.1073/pnas.0908286106
- Golovnev, A. V. (2017). Challenges to arctic nomadism: Yamal Nenets facing climate change era calamities. Arctic Anthropology, 54(2), 40–51. https://doi.org/10.3368/aa.54.2.40
- Golovnev, A. V., Lezova, S. V., Abramov, I. V., Belorussova, S. Y., & Babenkova, N. A. (2014). Ethno-expertise on Yamal: Nenets routes and gas fields. AMB Press. (in Russian).
- Habeck, J. O. (2007). What it means to be a Herdsman: The practice and image of Reindeer husbandry. Lit Verlag.
- Hegerl, G. C., Hoegh-Guldberg, O., Casassa, G., Hoerling, M., Kovats, S., Parmesan, C., et al. (2009). IPCC expert meeting on detection and attribution related to anthropogenic climate change the world meteorological organization good practice guidance paper on detection and attribution related to anthropogenic climate change good practice guidance paper o. In T. Stocker, C. Field, Q. Dahe, V. Barros, G.-K. Plattner, M. Tignor, et al. (Eds.), *IPCC expert meeting on detection and attribution related to anthropogenic climate change.*
- Istomin, K., & Dwyer, M. (2021). Reindeer herder's thinking: A comparative research on relations between economy, cognition, and way of life. Verlag der Kulturstiftung Sibirien.
- Jasiewicz, J., & Stepinski, T. F. (2013). Geomorphons A pattern recognition approach to classification and mapping of landforms. Geomorphology, 182, 147–156. https://doi.org/10.1016/j.geomorph.2012.11.005
- Jernsletten, J.-L. L., & Klokov, K. (2002). Sustainable Reindeer husbandry.
- Karger, D. N., Conrad, O., Böhner, J., Kawohl, T., Kreft, H., Soria-Auza, R. W., et al. (2017). Climatologies at high resolution for the Earth's land surface areas. *Scientific Data*, 4, 1–20. https://doi.org/10.1038/sdata.2017.122
- Karger, D. N., Conrad, O., Böhner, J., Kawohl, T., Kreft, H., Soria-Auza, R. W., et al. (2018). Climatologies at high resolution for the Earth's land surface areas [Dataset]. Dryad Digital Repository. https://doi.org/10.5061/dryad.kd1d4
- Kassambara, A. (2020). rstatix: Pipe-Friendly framework for basic statistical tests. Retrieved from https://cran.r-project.org/package=rstatix
- Klein, D. R. (1968). The introduction, increase, and crash of reindeer on St. Matthew island. Journal of Wildlife Management, 32(2), 350. https:// doi.org/10.2307/3798981
- Kolari, T. H. M., Kumpula, T., Verdonen, M., Forbes, B. C., & Tahvanainen, T. (2019). Reindeer grazing controls willows but has only minor effects on plant communities in Fennoscandian oroarctic mires. Arctic Antarctic and Alpine Research, 51(1), 506–520. https://doi.org/10.108 0/15230430.2019.1679940
- Kristensen, J. A., Svenning, J.-C., Georgiou, K., & Malhi, Y. (2022). Can large herbivores enhance ecosystem carbon persistence? Trends in Ecology & Evolution, 37(2), 117–128. https://doi.org/10.1016/j.tree.2021.09.006
- Kvashnin, I. N. (2009). The Nenets reindeer herding in the 20 c. Beginning of 21 c. (in Russian).
- Leibman, M. O. (1995). Cryogenic landslides on the Yamal Peninsula, Russia: Preliminary observations. Permafrost and Periglacial Processes, 6(3), 259–264. https://doi.org/10.1002/ppp.3430060307
- Lundgren, E. J., Ramp, D., Rowan, J., Middleton, O., Schowanek, S. D., Sanisidro, O., et al. (2020). Introduced herbivores restore Late Pleistocene ecological functions. *Proceedings of the National Academy of Sciences of the United States of America*, 117(14), 7871–7878. https://doi. org/10.1073/pnas.1915769117
- Macias-Fauria, M., Jepson, P., Zimov, N., & Malhi, Y. (2020). Pleistocene Arctic megafaunal ecological engineering as a natural climate solution? *Philosophical Transactions of the Royal Society B: Biological Sciences*, 375(1794), 20190122. https://doi.org/10.1098/rstb.2019.0122
- Meredith, M., Sommerkorn, M., Cassotta, S., Derksen, C., Ekaykin, A., Hollowed, A., et al. (2019). Polar regions. In H.-O. Pörtner, D. C. Roberts, V. Masson-Delmotte, P. Zhai, M. Tignor, E. Poloczanska, et al. (Eds.), *IPCC special report on the ocean and cryosphere in a changing climate.*

- Metcalfe, D. B., & Olofsson, J. (2015). Distinct impacts of different mammalian herbivore assemblages on arctic tundra CO₂ exchange during the peak of the growing season. *Oikos*, 124(12), 1632–1638. https://doi.org/10.1111/oik.02085
- Monteath, A. J., Gaglioti, B. V., Edwards, M. E., & Froese, D. (2021). Late Pleistocene shrub expansion preceded megafauna turnover and extinctions in eastern Beringia. Proceedings of the National Academy of Sciences of the United States of America, 118(52), e2107977118. https:// doi.org/10.1073/pnas.2107977118
- Myers-Smith, I. H., Elmendorf, S. C., Beck, P. S. A., Wilmking, M., Hallinger, M., Blok, D., et al. (2015). Climate sensitivity of shrub growth across the tundra biome. *Nature Climate Change*, 5(9), 887–891. https://doi.org/10.1038/nclimate2697
- Myers-Smith, I. H., Forbes, B. C., Wilmking, M., Hallinger, M., Lantz, T., Blok, D., et al. (2011). Shrub expansion in tundra ecosystems: Dynamics, impacts and research priorities. *Environmental Research Letters*, 6(4), 045509. https://doi.org/10.1088/1748-9326/6/4/045509
- Newton, E. J., Pond, B. A., Brown, G. S., Abraham, K. F., & Schaefer, J. A. (2014). Remote sensing reveals long-term effects of caribou on tundra vegetation. *Polar Biology*, 37(5), 715–725. https://doi.org/10.1007/s00300-014-1472-3
- Olofsson, J., Kitti, H., Rautiainen, P., Stark, S., & Oksanen, L. (2001). Effects of summer grazing by reindeer on composition of vegetation, productivity and nitrogen cycling. *Ecography*, 24(1), 13–24. https://doi.org/10.1034/j.1600-0587.2001.240103.x
- Olofsson, J., Oksanen, L., Callaghan, T., Hulme, P. E., Oksanen, T., & Suominen, O. (2009). Herbivores inhibit climate-driven shrub expansion on the tundra. *Global Change Biology*, 15(11), 2681–2693. https://doi.org/10.1111/j.1365-2486.2009.01935.x
- Olofsson, J., & Post, E. (2018). Effects of large herbivores on tundra vegetation in a changing climate, and implications for rewilding. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 373(1761), 20170437. https://doi.org/10.1098/rstb.2017.0437
- Olofsson, J., Stark, S., & Oksanen, L. (2004). Reindeer influence on ecosystem processes in the tundra. *Oikos*, 105(2), 386–396. https://doi.org/10.1111/j.0030-1299.2004.13048.x
- Olofsson, J., Tommervik, H., & Callaghan, T. V. (2012). Vole and lemming activity observed from space. Nature Climate Change, 2(12), 880–883. https://doi.org/10.1038/nclimate1537
- Pajunen, A. M. (2009). Environmental and biotic determinants of growth and height of Arctic willow shrubs along a latitudinal gradient. Arctic Antarctic and Alpine Research, 41(4), 478–485. https://doi.org/10.1657/1938-4246-41.4.478
- Parker, T. C., Thurston, A. M., Raundrup, K., Subke, J., Wookey, P. A., & Hartley, I. P. (2021). Shrub expansion in the Arctic may induce large-scale carbon losses due to changes in plant-soil interactions. *Plant and Soil*, 463(1–2), 643–651. https://doi.org/10.1007/s11104-021-04919-8
- Parmesan, C., Burrows, M. T., Duarte, C. M., Poloczanska, E. S., Richardson, A. J., Schoeman, D. S., & Singer, M. C. (2013). Beyond climate change attribution in conservation and ecological research. *Ecology Letters*, 16, 58–71. https://doi.org/10.1111/ele.12098
- Pauly, D. (1995). Anecdotes and the shifting baseline syndrome of fisheries. Trends in Ecology & Evolution, 10(10), 430. https://doi.org/10.1016/ S0169-5347(00)89171-5
- Porter, C., Morin, P., Howat, I., Noh, M.-J., Bates, B., Peterman, K., et al. (2018). ArcticDEM. Harvard Dataverse, V1. https://doi.org/10.7910/ DVN/OHHUKH
- Post, E., & Pedersen, C. (2008). Opposing plant community responses to warming with and without herbivores. Proceedings of the National Academy of Sciences of the United States of America, 105(34), 12353–12358. https://doi.org/10.1073/pnas.0802421105
- Ravolainen, V. T., Bråthen, K. A., Ims, R. A., Yoccoz, N. G., Henden, J.-A., & Killengreen, S. T. (2011). Rapid, landscape scale responses in riparian tundra vegetation to exclusion of small and large mammalian herbivores. *Basic and Applied Ecology*, 12(8), 643–653. https://doi. org/10.1016/j.baae.2011.09.009
- Raynolds, M. K., Walker, D. A., Balser, A., Bay, C., Campbell, M., Cherosov, M. M., et al. (2019). A raster version of the circumpolar Arctic vegetation map (CAVM). *Remote Sensing of Environment*, 232(January), 111297. https://doi.org/10.1016/j.rse.2019.111297
- Raynolds, M. K., Walker, D. A., Epstein, H. E., Pinzon, J. E., & Tucker, C. J. (2012). A new estimate of tundra-biome phytomass from trans-Arctic field data and AVHRR NDVI. *Remote Sensing Letters*, 3(5), 403–411. https://doi.org/10.1080/01431161.2011.609188
- R Core Team. (2020). R: A language and environment for statistical computing. Retrieved from https://www.r-project.org/
- Rees, W. G., Williams, M., & Vitebsky, P. (2003). Mapping land cover change in a reindeer herding area of the Russian Arctic using Landsat TM and ETM+ imagery and indigenous knowledge. *Remote Sensing of Environment*, 85(4), 441–452. https://doi.org/10.1016/S0034-4257(03)00037-3
- Ricca, M. A., Miles, A. K., Van Vuren, D. H., & Eviner, V. T. (2016). Impacts of introduced Rangifer on ecosystem processes of maritime tundra on subarctic islands. *Ecosphere*, 7(3), e01219. https://doi.org/10.1002/ecs2.1219
- Rickbeil, G. J. M., Coops, N. C., & Adamczewski, J. (2015). The grazing impacts of four barren ground caribou herds (*Rangifer tarandus* groenlandicus) on their summer ranges: An application of archived remotely sensed vegetation productivity data. *Remote Sensing of Environment*, 164, 314–323. https://doi.org/10.1016/j.rse.2015.04.006
- Running, S., Mu, Q., Zhao, M., & Moreno, A. (2019). MOD16A2GF MODIS/Terra net evapotranspiration gap-filled 8-day L4 global 500 m SIN grid V006 [Dataset]. NASA EOSDIS Land Processes DAAC. https://doi.org/10.5067/MODIS/MOD16A2GF.006
- Seddon, A. W. R., Macias-Fauria, M., Long, P. R., Benz, D., & Willis, K. J. (2016). Sensitivity of global terrestrial ecosystems to climate variability. *Nature*, 531(7593), 229–232. https://doi.org/10.1038/nature16986
- Skarin, A., Verdonen, M., Kumpula, T., Macias-Fauria, M., Alam, M., Kerby, J., & Forbes, B. C. (2020). Reindeer use of low Arctic tundra correlates with landscape structure. *Environmental Research Letters*, 15(11), 115012. https://doi.org/10.1088/1748-9326/abbf15
- Spiegel, M. P., Volkovitskiy, A., Terekhina, A., Forbes, B. C., Park, T., & Macias-Fauria, M. (2022a). marcusspiegel/yamal_rs: yamal_rs v1.0.0. https://doi.org/10.5281/zenodo.7387883
- Spiegel, M. P., Volkovitskiy, A., Terekhina, A., Forbes, B. C., Park, T., & Macias-Fauria, M. (2022b). Top-down regulation by a reindeer herding system limits climate-driven Arctic vegetation change at a regional scale: Data. Oxford University Research Archive. https://doi.org/10.5287/ bodleian:E2x6bnggk
- Stammler, F. (2005). Reindeer nomads meet the market: Culture, property and globalisation at the "End of the land". Lit.
- Sturm, M., Schimel, J., Michaelson, G., Welker, J. M., Oberbauer, S. F., Liston, G. E., et al. (2005). Winter biological processes could help convert arctic tundra to shrubland. *BioScience*, 55(1), 17. https://doi.org/10.1641/0006-3568(2005)055[0017:WBPCHC]2.0.CO;2
- Väisänen, M., Ylänne, H., Kaarlejärvi, E., Sjögersten, S., Olofsson, J., Crout, N., & Stark, S. (2014). Consequences of warming on tundra carbon balance determined by reindeer grazing history. *Nature Climate Change*, 4(5), 384–388. https://doi.org/10.1038/nclimate2147
- van der Wal, R. (2006). Do herbivores cause habitat degradation or vegetation state transition? Evidence from the tundra. *Oikos*, 114(1), 177–186. https://doi.org/10.1111/j.2006.0030-1299.14264.x
- Verma, M., Schulte to Bühne, H., Lopes, M., Ehrich, D., Sokovnina, S., Hofhuis, S. P., & Pettorelli, N. (2020). Can reindeer husbandry management slow down the shrubification of the Arctic? *Journal of Environmental Management*, 267(April), 110636. https://doi.org/10.1016/j. jenvman.2020.110636
- Veselkin, D. V., Morozova, L. M., & Gorbunova, A. M. (2021). Decrease of NDVI values in the southern tundra of Yamal in 2001–2018 correlates with the size of domesticated reindeer population. Sovremennye Problemy Distantsionnogo Zondirovaniya Zemli Iz Kosmosa, 18(2), 143–155. https://doi.org/10.21046/2070-7401-2021-18-2-143-155

- Volkovitskiy, A., & Terekhina, A. (2020). The contemporary issues of Yamal reindeer herding: Discussions and perspectives. *Etnografia*, 2(8), 152–169. (in Russian).
- von Eye, A., Mun, E., & Mair, P. (2012). Log-linear modeling. WIREs Computational Statistics, 4(2), 218–223. https://doi.org/10.1002/wics.203 Walker, D. A., Leibman, M. O., Epstein, H. E., Forbes, B. C., Bhatt, U. S., Raynolds, M. K., et al. (2009). Spatial and temporal patterns of green-
- ness on the Yamal Peninsula, Russia: Interactions of ecological and social factors affecting the Arctic normalized difference vegetation index. Environmental Research Letters, 4(4), 045004. https://doi.org/10.1088/1748-9326/4/4/045004
- Walker, D. A., Raynolds, M. K., Daniëls, F. J. A., Einarsson, E., Elvebakk, A., Gould, W. A., et al. (2005). The circumpolar Arctic vegetation map. Journal of Vegetation Science, 16(3), 267–282. https://doi.org/10.1111/j.1654-1103.2005.tb02365.x
- Ylänne, H., Kaarlejärvi, E., Väisänen, M., Männistö, M. K., Ahonen, S. H. K., Olofsson, J., & Stark, S. (2020). Removal of grazers alters the response of tundra soil carbon to warming and enhanced nitrogen availability. *Ecological Monographs*, 90(1), 1–13. https://doi.org/10.1002/ ecm.1396
- Ylänne, H., Olofsson, J., Oksanen, L., & Stark, S. (2018). Consequences of grazer-induced vegetation transitions on ecosystem carbon storage in the tundra. *Functional Ecology*, 32(4), 1091–1102. https://doi.org/10.1111/1365-2435.13029
- Zimov, S. A., Zimov, N. S., Tikhonov, A. N., & Chapin, I. S. (2012). Mammoth steppe: A high-productivity phenomenon. Quaternary Science Reviews, 57, 26–45. https://doi.org/10.1016/j.quascirev.2012.10.005